

# **Geophysical Research Letters**®

# **RESEARCH LETTER**

10.1029/2021GL094339

I. Vidaller and J. Revuelto are both cofirst authors.

### **Key Points:**

- The glacierized area shrank by 23.2% and thickness decreased on average by 6.3 m during the 2011–2020 period
- There is no sign of slowdown in glacier shrinkage respect to previous decades
- The smaller Pyrenean glaciers are influenced by local topography as they show highly contrasted evolution under the same climatic conditions

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. Revuelto, jrevuelto@ipe.csic.es

#### **Citation**:

Vidaller, I., Revuelto, J., Izagirre, E., Rojas-Heredia, F., Alonso-González, E., Gascoin, S., et al. (2021). Toward an ice-free mountain range: Demise of Pyrenean glaciers during 2011– 2020. *Geophysical Research Letters*, *48*, e2021GL094339. https://doi. org/10.1029/2021GL094339

Received 18 MAY 2021 Accepted 17 AUG 2021

#### © 2021 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

#### VIDALLER ET AL.

# Toward an Ice-Free Mountain Range: Demise of Pyrenean Glaciers During 2011–2020

I. Vidaller<sup>1</sup> <sup>(D)</sup>, J. Revuelto<sup>1</sup> <sup>(D)</sup>, E. Izagirre<sup>2,3</sup> <sup>(D)</sup>, F. Rojas-Heredia<sup>1</sup> <sup>(D)</sup>, E. Alonso-González<sup>1</sup> <sup>(D)</sup>, S. Gascoin<sup>4</sup> <sup>(D)</sup>, P. René<sup>5</sup> <sup>(D)</sup>, E. Berthier<sup>6</sup> <sup>(D)</sup>, I. Rico<sup>7</sup> <sup>(D)</sup>, A. Moreno<sup>1</sup> <sup>(D)</sup>, E. Serrano<sup>8</sup> <sup>(D)</sup>, A. Serreta<sup>9</sup>, and J. I. López-Moreno<sup>1</sup> <sup>(D)</sup>

<sup>1</sup>Instituto Pirenaico de Ecología, Consejo Superior de Investigaciones Científicas (IPE-CSIC), Zaragoza, Spain, <sup>2</sup>Department of Geology, University of the Basque Country UPV/EHU, Leioa, Spain, <sup>3</sup>Basque Centre for Climate Change BC3, Leioa, Spain, <sup>4</sup>Centre d'Etudes Spatiales de la Biosphère, CESBIO, Université de Toulouse, CNES/ CNRS/INRAE/IRD/UPS, Toulouse, France, <sup>5</sup>MORAINE Association Pyrénéenne de Glaciologie, Luchon, France, <sup>6</sup>LEGOS, Université de Toulouse, CNES, CNRS, IRD, UPS, Toulouse, France, <sup>7</sup>Department of Geography, Prehistory, and Archaeology, University of the Basque Country UPV/EHU, Vitoria-Gasteiz, Spain, <sup>8</sup>Department of Geography, GIR PANGEA, University of Valladolid, Valladolid, Spain, <sup>9</sup>Department of Design and Manufacturing Engineering, University of Zaragoza, Zaragoza, Spain

**Abstract** Pyrenean glaciers are the largest in southern Europe. Their survival is threatened by climate change, highlighting the significance of their study. This research presents an assessment of changes in the glacierized area and thickness of Pyrenean glaciers from 2011 to 2020, using high-resolution optical satellite, airborne lidar and UAV images. The total glacierized area has shrunk by 23.2% and thickness has decreased on average by 6.3 m. These two variables show no correlation for individual glaciers. Although climatic conditions do not vary much among glaciers, their evolution was heterogeneous during the study period. The smaller glaciers (<10 ha) show a higher variability in their area decrease and thickness loss whereas the four largest glaciers (>10 ha) have a more homogeneous response. This can be attributed to the generally larger influence of local topography on the response of the smaller Pyrenean glaciers. There is no sign of slowdown in glacier shrinkage respect to previous decades.

**Plain Language Summary** Pyrenean glaciers are the largest in southern Europe. Their survival is threatened by climate change, highlighting the significance and relevance of their study. This study presents an assessment of changes in the glacierized area and thickness of Pyrenean glaciers during 2011–2020 based on UAV, optical satellite imagery and lidar observations. In this period, their total area shrank by 23.2% and thickness decreased by 6.3 m on average. Although climatic conditions do not vary much among glaciers, their evolution was heterogeneous during the observed period. The smaller Pyrenean glaciers (area <10 ha) are highly controlled by local topography, whereas the largest glaciers are predominantly influenced by regional climate forcing. There is no sign of slowdown in shrinkage of Pyrenean glaciers respect to previous decades. This indicates the continuous decline of Pyrenean glaciers toward an ice-free mountain range in the coming decades.

# 1. Introduction

Glacier mass balance evolution depends on snow accumulation and snow, firn and ice melt during the cold and warm seasons respectively and is thus considered a reliable indicator of climate fluctuations (Braithwaite & Hughes, 2020). The Little Ice Age (LIA, between the 14th and 19th centuries) represents the last global advance phase for the majority of mountain glaciers around the world (Solomina et al., 2016). Since then, the decline of glaciers has been almost continuous, only interrupted by short stabilization periods (Zemp et al., 2015). Several studies identify the 1980s as a "tipping point in global glacier evolution," followed by accelerated glacier shrinkage (Beniston et al., 2018; Huss & Hock, 2018).

Very small glaciers (<0.5 km<sup>2</sup>) predominate in number in the northern hemisphere mountain ranges at temperate latitudes, since more than 80% of glaciers in these mountains are beneath this area threshold (Huss & Fischer, 2016). Shrinkage of very small glaciers occurred more rapidly by the late 20<sup>th</sup> and early  $21^{st}$  centuries than in earlier decades (Bahr & Radić, 2012; Parkes & Marzeion, 2018). This fast shrinkage is explained by their generally low accumulation area ratio, which is mainly driven by the observed global

temperature increase (Zemp et al., 2015). However, some of these very small glaciers, when constrained to the most elevated and sheltered areas, start to attenuate their response to regional climate evolution, and local topographic factors influence their mass balance, leading to the so-called topoclimatic control (Fischer, 2018; Florentine et al., 2018, 2020; Huss & Fischer, 2016). Under these circumstances a small fraction of very small glaciers still persist at relatively low elevation, often below the regional equilibrium-line altitude (ELA; Grunewald et al., 2006; Hughes & Woodward, 2009), demonstrating that they are decoupled from regional climate forcing (Fischer, 2018).

The southernmost glaciers in Europe are located in the Pyrenees, Apennines and Balkan Peninsula at latitudes between 41°N and 43°N and elevations between 2,000 and 3,300 m above sea level (a.s.l.) (Grunewald & Scheithauer, 2010) and can all be classified as very small glaciers as all of them are smaller than 0.5 km<sup>2</sup>. Since the LIA maximum during the 19th century, these glaciers have lost 30%–100% of their volume (Hughes, 2018). In southern Europe, glaciers larger than 0.05 km<sup>2</sup> (5 ha) are only found in the Pyrenees. The Pyrenean glaciers are in extreme jeopardy and could disappear or become residual ice patches in about 20 years from now (López-Moreno, García-Ruiz et al., 2020). This kind of process has already been observed in other southern European mountain ranges (Carturan et al., 2013) or in tropical areas such as the Santa Isabel Mountains (Colombia) and the Puncak Jaya Mountains (Indonesia) where glacier extinction is imminent (Morán-Tejeda et al., 2018).

In the Pyrenees, temperature has increased by more than 1.5°C since the LIA (Cuadrat et al., 2018). For high-elevation areas, this warming has been especially marked during the summer and shoulder seasons (López-Moreno et al., 2019). Snow depth observations in the Spanish Pyrenees started in mid 1980s and exhibit a significant decline, especially in spring at elevations above 2,000 m a.s.l. (López-Moreno, 2005). Analyses of long-term (1958–2017) trends in snow cover duration and depth by López-Moreno, Soubeyroux et al. (2020) confirmed the significant decrease in snow depth especially at high elevation sites of the French and Spanish Pyrenees during spring time. These trends led to a marked area shrinkage of the Pyrenean glaciers, from 2,060 ha in 1,850 to 810.3 ha in 1984 (a loss of 60.7%) to 242.0 ha in 2016 (a loss of 88.3% compared to 1,850) and the number of glaciers has decreased accordingly, from 52 (in 1850) to 39 (1984) to 19 (2016) (according to Rico et al., 2017). The main question to be answered about the evolution of Pyrenean glaciers is whether they are still controlled by regional climatic conditions and still show accelerated shrinkage and wastage, or whether glacier changes are increasingly controlled by topographic factors that might attenuate their response to temperature increase. Whereas the latter behavior has been observed for glaciers in other Mediterranean regions (Hughes, 2018), the attempt to answer this question for the Pyrenees has been hampered by the scarcity of mass balance/volume estimations available to date, since there are only observations from Ossoue (René, 2014), Maladeta (Pastor-Argüello, 2013) and Monte Perdido (López-Moreno et al., 2019) glaciers for limited time periods.

In this study, we report and analyse glacier area, surface elevation and thickness changes for Pyrenean glaciers from 2011 to 2020. Glacier area changes have been determined for the 24 remaining Pyrenean glaciers in 2011 from high spatial resolution satellite imagery and field surveys. Additionally, glacier volume changes have been determined from 3D point clouds derived from close-range remote-sensing techniques for 17 of the 24 remaining glaciers. This data set contributes to identifying the most important driving factors of recent glacier evolution all along the Pyrenean mountain range and, hence, determining whether these glaciers are responding homogeneously to climatic changes or whether topographic factors are significantly affecting and influencing their response.

# 2. Data and Methods

#### 2.1. Study Area

Glaciers in the Pyrenees are located only in the central and highest massifs. Glaciers of the *Infiernos, Monte Perdido, Posets* and *Maladeta* massifs are located in Spain, while glaciers in *Balaitous, Vignemale, Gavarnie, La Munia, Perdiguero* and *Mont Valier* massifs are in France (Figure S1). The estimated mean annual elevation of the 0°C isotherm ranges between 2,700 and 3,000 m a.s.l. (Jomelli et al., 2020). At the highest elevations, the mean annual precipitation exceeds 2,000 mm, winter and spring being the most humid seasons (Buisan et al., 2015).

# 2.2. Computation of Glacier Thickness Change

#### 2.2.1. Acquisitions of Consecutive Glacier Surface Elevation Information

Glacier surface elevation data were obtained from different close-range remote-sensing devices. In 2011, glacier surface elevation information was acquired with airborne lidar, known to produce high-resolution input data for glacier surface elevation changes (Hopkinson & Demuth, 2006). Between October and November 2011 (October 17–21 for western parts of the study area and November 4–9 for eastern parts of the study area) the Spanish National Geographic Institute (IGN, last accessed May 2021: http://centrode-descargas.cnig.es/CentroDescargas/index.jsp) conducted several airborne lidar flights over the Pyrenees to generate a high spatial resolution Digital Elevation Model (DEM). The instrument used was a Leica ALS60, which has a diode-pumped transmitter and a large-aperture low-inertia/high-speed scan mirror working with a wavelength of 1,064 nm. The airborne lidar surveys generated point clouds with an average density of 0.5 pts/m<sup>2</sup>.

In 2020, glacier surface elevation information was obtained with three different UAVs (according to the logistic requirements of the sites: two fixed-wing UAVs (*SenseFly eBee-Plus* and *eBee-X*) and a quad-copter UAV [*DJI Mavic Pro 2*]; Table S1) in several field campaigns between late August and early October. The fixed-wing UAVs are equipped with a *SenseFly* S.O.D.A. digital camera (20Mp resolution), with GPS receivers allowing RTK (Real Time Kinematic) or PPK (Post-Processed Kinematic) positioning systems (positioning accuracy <0.05 m). The quad-copter also has a camera of 20Mp resolution and a GPS with a positioning accuracy >5 m.

The overlaps of the UAV images (always higher than 70%), together with the ground sampling distance (GSD) of about 2.8 cm/px defined by the UAV flight elevation, allowed to generate dense 3D point clouds of the study areas using the Structure from Motion (SfM) algorithm (Forlani et al., 2018). The SfM software used to process all images and create the point clouds was *Pix4Dmapper (Pix4D)*. The images acquired and the Pix4D point cloud densification algorithms allowed generation of 3D point clouds with an average resolution of 4 pts/m<sup>2</sup>.

The *Monte Perdido* glacier surface was surveyed in 2011 and 2020 (both in early October) with a long-range Terrestrial Laser Scanner (TLS). The device, a RIEGL LPM-321, uses lidar time-of-flight (at 905 nm wave-length) to generate a 3D point cloud (Revuelto et al., 2014). Details of the acquisition procedure and point cloud generation with this device can be found in López-Moreno et al. (2019).

## 2.2.2. Point Cloud Geolocation and Glacier Thickness Change Computation

The images obtained with the two fixed-wing UAVs allowed an accurate geolocation due to the RTK-PPK GPS options. By contrast, in the case of quad-copter UAVs, images were geolocated with the standard GPS geolocation error (~5 m), and thus the point clouds generated with this UAV had a lower 3D positional accuracy. These point clouds were co-registered with the airborne lidar point clouds in *Cloud Compare* (Girardeau-Montaut, 2016) to allow the later assessment of glacier surface elevation changes. First, several areas of stable terrain around the glaciers (at least seven ice-free patches covering all directions) with no expected changes in surface elevation during the study period (ridge, peaks, etc.,) were selected in both point clouds to align the UAV to the lidar point cloud. Afterward, these areas of stable terrain were used to compute a rotation and translation matrix with the iterative closest point algorithm (Rajendra et al., 2014). Finally, this matrix was applied to the entire point clouds of the quad-copters (see the Root Mean Squared Error (RMSE) values in Table S1).

Point clouds differences over glacierized areas (glacier thickness change) between 2011 and 2020 point clouds were obtained with the *M3C2* tool (James, 2017) of *Cloud Compare*. Then, these differences were rasterized for further analysis. Glacier surface elevation changes was calculated over the glacierized area existing in 2020. Readers are referred to the "Uncertainty of the changes observed" included in the Supporting Information S1.





**Figure 1.** (a) Distribution of glacier elevations in 2011 categorized by aspect. (b) Distribution of thickness losses by aspect. The size of each sector on upper graphs corresponds to the percentage of the total glaciated area. The color scales show entire elevation range (a) or thickness loss (b and c) Altitudinal distribution of relative area loss (light green bars) and glaciated area in 2011 (blue points) and 2020 (pink points). (d) Distribution of relative area loss and glaciated area in 2011 and 2020.

#### 2.3. Glacier Area Delimitation

Glacier delimitation was made using manual on screen tracing of ice margins in both 2011 and 2020. For 2011, glacier outlines were delimited from RapidEye satellite images, which were acquired between September and October and validated against the glaciers outlines from Marti, Gascoin, Houet, Ribière, et al., 2015 and Marti, Houet et al., 2015. For 2020 glacier outlines, high-resolution imagery derived from the UAVs and field survey with a handheld GPS was used for the manual delimitation of glaciers boundaries. For both datasets of glacier outlines (2011 and 2020), the projected glacier surface in a 2D horizontal plane and the 3D representation were computed. For the latter, glacier extents were calculated from the 3D UAV, the TLS or lidar surveys of the glaciers' surface topography. Information about the uncertainty of glacier area delimitation is included in the Supporting Information S1: "Uncertainty of the changes observed."

# 3. Results and Discussion

Based on high spatial resolution remote-sensing imagery from satellites and field surveys, 24 glaciers were mapped for 2011, covering a 2D area of 293.9 ha (2.93 km<sup>2</sup>) which corresponds to a 3D surface of 338.9 ha (3.39 km<sup>2</sup>) (hereafter called 3D area; Figure 1; Table S2). In 2020, three ice bodies (*Coronas, Pailla East* and

*Maupas*) showed no sign of movement or their surface area was smaller than 2 ha, so these were not considered to be glaciers anymore but ice patches (Leigh et al., 2019). Several very small glaciers showed very weak movement features in 2020 (e.g., *La Paul, Tempestades*), symptomatic of their close demise. One of the three largest glaciers (*Aneto*) was divided into two separate ice bodies during the study period. The total glacierized area was reduced to 229.2 ha (257 ha 3D area), representing a loss of 23.2% (24.2% 3D area) in the 9 years of the study period, i.e., an average area loss of 2.6% per year. The area-weighted mean ice thickness loss was  $6.3 \pm 0.4$  m (0.70 m yr<sup>-1</sup>), equivalent to a specific mass balance loss rate of 0.59 m w. e. yr<sup>-1</sup> (Table S2), assuming a density conversion factor of 850 kg/m<sup>3</sup> (Huss, 2013). Many glaciers have shown zones where ice thickness decreased by more than 20 m. Despite these marked changes in glacier thickness the glacierized surface hypsography did not show strong variations, during the nine years of the study period.

The RMSE of glacier thickness change (Table S1 and Figure S4) shows that confidence in the ice loss estimation is high: in all cases, the RMSE is at least one order of magnitude lower than the mean glacier thickness change. However, some glaciers show higher uncertainties, mainly because of the lack of fixed common ground control points during glacier surface elevation acquisitions (in 2011 lidar flights and in 2020 in the UAV), potential surface elevation changes around glaciated areas being highly difficult to identify (debris falls, buried ice melting, etc.,), and in some glaciers the presence of a shallow snowpack (Fischer et al., 2016). On average, the residuals outside the glaciers (Figure S5) are below 0.4 m (accepted here as the uncertainty of thickness loss, see Supporting Information S1), an order of magnitude smaller than glacier-wide surface elevation changes which exceeded 6 m. This uncertainty (0.4 m) is assigned to the comparison of the two acquisition techniques as hardly any changes in surface elevation over stable terrain occurred during the observed period.

In 2011, the mean orientation of Pyrenean glaciers was concentrated between N and NE aspects (77% of the total), and their mean elevation above 2700 m a.s.l. (Figure 1a). A similar pattern persists in 2020 as N and NE aspects account for 60% of the glacierized area. However, glaciers with N aspect (respectively NE aspect) exhibited a relative area reduction of 22% (resp. 33%) from 2010 to 2020. In 2020, N and NNE aspect classes concentrate the highest fraction of glacierized area, and also show the largest thickness losses (>20 m at some locations, Figure 1b).

Below 2,400 m a.s.l., glacier area did not change significantly. At elevations from 2,400 to 2,700 m a.s.l., glaciers exhibited little shrinkage (~10%), whereas glacier area shrinkage was much higher (>30%) between 2800 and 3,000 m a.s.l. (Figure 1c). Such difference in glacier surfaces shrinkage is causing an increase in the average glacier slopes (López-Moreno et al., 2019). This could cause a decrease in snow accumulation, which is known to decline over 25–30° slopes due to gravitational displacement of fresh fallen snow (López-Moreno et al., 2017). Considering aspects, the largest relative area changes occurred in the SW aspect (-50%), while the largest absolute area losses were observed for the aspect class N. This is because on S slopes glaciers covered only a few hectares, so small area changes led to large relative losses (Figure 1d).

In Figure 2, the lack of a clear relation between ice thickness loss and loss of glacierized area for each individual glacier is striking. Nor is any clear relationship evident between glacier shrinkage and glacier aspect, mean elevation or size in 2011, also glacier hypsography did not change significantly between 2011 and 2020. Only the four largest Pyrenean glaciers (larger than 10 ha in 2020: Aneto, Ossoue, Monte Perdido and Maladeta, which represent 60.2% of the total 2020 area) show a positive correlation between area decrease and thickness loss. Ossoue, which shows an ESE aspect and a moderate mean elevation (3,068 m a.s.l), experienced the largest losses (25.7% area change and 10.9 m average thickness change), followed by Aneto (24.4% and 8.5 m) and Maladeta glaciers (18.4% and 7.4 m). Monte Perdido exhibited reduced area losses but similar ice thickness loss (12.9% and 7.5 m). For three out of this four glaciers, yearly surface mass balances (Lopez-Moreno et al., 2019; Pastor-Argüello 2013; René, 2014) show that these glaciers are mainly controlled by the mean climatic conditions of the year (snow accumulation in winter and air temperatures in summer). Smaller glaciers (area <10 ha) exhibited strongly contrasting area and thickness loss. Among these latter glaciers, three classes can be distinguished. A first group of glaciers (4-10 ha) surrounded by rock walls facing N and NE, shows small area and thickness loss, including Infierno, Oulettes, Tempestades, and La Paul glaciers, characterized by very heterogeneous local conditions (summit altitudes and ice thickness, height of rock walls, lithology). These glaciers are also the least dynamic glaciers. The second class also corresponds to glaciers in the size range of 4-10 ha but with very high thickness loss and low area loss (Seil





**Figure 2.** Relationship between thickness loss (m) and glacierized area reduction (%). Triangle size represents 2011 area, the color scale represents the mean glacier elevation and the orientation of the triangles refers to the mean aspect of the glacier.

*de la Baque, Taillon, Gabietous*, and *Llardana*). These glaciers have moderate elevations but are situated in narrow cirques surrounded by nearly vertical rock walls with distinct main aspect classes and thus local topographical conditions are decisive for their strong mass loss. The third class is formed by the smallest Pyrenean glaciers (<4 ha) and ice patches located at high elevation with high area loss and moderate thickness loss, located in open morphology (i.e., not protected against solar radiation by narrow cirques or rock walls) with moderately steep rocks walls in granitic massifs.

These results highlight the marked variability in thickness loss among glaciers. Figures 3, S2 and S3 illustrate that in most cases the maximum thinning is found in the lower-lying and middle part of the glaciers, while the upper parts exhibit an attenuated wastage or even elevation changes close to zero (the largest reductions in ice thickness are observed at the elevations with the largest percentage of total glacier cover). Glaciers that are split into two different ice bodies (i.e., *Monte Perdido* and *Taillon*) show a bimodal distribution of ice thickness loss with elevation.

Although Pyrenean glaciers have already retreated to the most elevated and protected cirques (Rico et al., 2017), they are still in clear imbalance with the current climate in the Pyrenees. As this work shows, the reach of a new equilibrium that could be expected in the evolution of very small and marginal glaciers (Carturan et al., 2013) has not yet occurred in the Pyrenees. This may cause the complete disappearance of most Pyrenean glaciers in the coming decades. Indeed, the loss of 23.2% (2.60% per year) of the glacierized area in the period 2011–2020 is similar to the rates of loss of 2.93% per year and 2.59% per year reported for the periods 2008–2016 (Rico et al., 2017) and 1984–2008 (Rico, 2019), respectively. The mass balances of -0.70 and -1.03 m w. e. yr<sup>-1</sup> for *Maladeta* and *Ossoue* glaciers do not differ substantially from previously reported mass balances (Gascoin & René, 2018; Marti, Gascoin, Houet, Laffly, et al., 2015). Also, the area-weighted mean ice thickness loss of 0.70 m yr<sup>-1</sup> reported here is somewhat smaller but in the same order of magnitude compared to values reported by Hugonnet et al. (2021) in Central Europe ( $-1.00/-1.11 \pm 0.23/0.26$  m





**Figure 3.** Lower panel: 3D views depict surface elevation change for the three largest glaciers in the Pyrenees during 2011–2020 (from left to right: *Ossoue, Monte Perdido* and *Aneto*). The black line in the three maps delineates the 2020 glacier area. Upper panel: boxplots show the mean surface elevation change for each elevation band; the pink line shows the distribution of glacier area with elevation. Since the 3D views show a frontal view of glaciers, each image has in the upper part an arrow indicating the North direction.

 $yr^{-1}$  in the periods 2010–2014/2015–2019, respectively for a much wider region). Nonetheless, high spatial resolution observations of glacier evolution presented here show differences when compared with coarser-resolution studies (Fischer et al., 2016). Thus, the surface elevation change reported in two 1° × 1° grids for the Pyrenees glaciers (42°N, 0°E and 42°N, 1°W) shows a mean ice thickness change of -0.96 m yr <sup>-1</sup> over a long time period (2000–2019; considering the glaciated area inside each 1° size pixel of the Pyrenees; Hugonnet et al., 2021), contrasting with the -0.70 m yr<sup>-1</sup> observed here. Despite the uncertainties in both works (0.07 and 0.04 m yr<sup>-1</sup> respectively) and also the differing study periods, the difference in mean values is not negligible. This shows the importance of in-situ observations, which are needed to correct glacier elevation or ice thickness changes derived by coarser satellite imagery.

Temperature and precipitation anomalies from ERA5 reanalysis data (Hersbach et al., 2020) for the study period 2011–2020, compared to a reference period from 1980 to 2010 (total precipitation from November to May and average temperature from June to September), showed that precipitation was slightly higher (+9%) and temperature warmer (+0.7°C on average) during the study period (Figure S4). However, these changes in climatic conditions were similar for all glaciers and their response to these variations was highly variable. A strong variability in area and thickness loss is observed among the smallest glaciers. This variable response to similar climatic conditions strongly suggests that local topographic conditions are becoming more important and climate evolution may have a lesser impact on glacier dynamics (Fischer, 2018; Hughes, 2018; Huss & Fischer, 2016; Rico, 2019). Moreover, the variability in glacier area decrease and thickness loss is more pronounced among the smallest Pyrenean glaciers which, probably highlights that, the smaller a very small glacier becomes, the higher the influence of local topography on its evolution. Recent glacier changes are related to the cirque characteristics of each glacier, such as shadowing from neighboring slopes, glacier slope, position relative to dominant blowing snow direction, and snow avalanche deposition. This result agrees with previous studies that point out that small glaciers have an exacerbated spread in response to regional climate fluctuations (DeBeer & Sharp, 2009; Fischer et al., 2014, 2015). Conversely, the four largest

Pyrenean glaciers exhibited changes more consistent with each other, with a much smaller range of variability in area and thickness losses than when all ice bodies are compared to the entire sample of Pyrenean glaciers. This suggests that the evolution of these four glaciers are still primarily influenced by the regional climate signal. However, some differences in the behavior of these larger glaciers appear, such as the larger losses of ice of *Ossoue* glacier (10.9 m of mean thickness loss and 25.7% of area reduction) compared to *Monte Perdido* glacier (7.5 m mean thickness loss and 12.9% area decrease) that, could be explained by their highly contrasted potential incoming solar radiation (respectively, mean daily radiation of 0.17 and 0.12 kW  $m^{-2}$  from early June to late September). *Ossoue* glacier is located inside a topographic depression where there are no large slopes, hence cirgue walls do not protect it from direct solar radiation.

Glacierized area losses and thickness loss for the different glaciers are highly heterogeneous, confirming that glacier area cannot be used as a proxy for mass balance (Huss & Fischer, 2016). As recent research and the results of this study demonstrate, TLS surveys (Fischer et al., 2016; López-Moreno et al., 2019), UAV surveys (Gaffey & Bhardwaj, 2020; Revuelto et al., 2021) and new emerging remote-sensing products such as high resolution satellite stereo imagery (Gascoin & René, 2018) are efficient tools for analyzing the evolution of very small glaciers. The increased use of optical satellite imagery with enhanced resolution and frequency may permit better delimitation of glacierized areas, and the identification of crevasses, debris cover and the evolution of the ELA, finally allowing an improved monitoring of even very small glaciers (Marti, Gascoin, Houet, Laffly, et al., 2015).

Available estimations from ground-penetrating radar surveys suggest that the maximum ice thickness of Pyrenean glaciers rarely exceeds 30 m and most of the glacierized areas are less than 15 m thick (Del Rio et al., 2014; Jiménez-Vaquero, 2016; López-Moreno et al., 2019). When these values are contrasted to ice losses reported in this study for the last 9 years, we can argue with confidence that Pyrenean glaciers are in extreme jeopardy and could disappear or become residual ice patches in about two decades.

## 4. Conclusions

From 2011 to 2020, the glacierized area of the Pyrenees has been reduced from 293.9 to 229.2 ha (-23.2%)and three glaciers have disappeared. The area-weighted mean ice thickness loss was 6.3 m, which is equivalent to an area-weighted specific mass balance of -0.59 m w. e. yr<sup>-1</sup>. The annual rate of glacierized area loss reported in the study period is similar to that observed since the 1980s. This indicates that glacier shrinkage in the Pyrenees and wastage have not slowed down in the last few years. However, the individual values of glacier shrinkage and area reduction vary significantly among glaciers. Moreover, these two variables are poorly correlated, suggesting that change in surface area is a poor proxy for mass balance in this region. The four largest Pyrenean glaciers have exhibited strong losses in area and ice thickness, causing important changes in their slope and mean elevation. The other glaciers (those covering less than 10 ha) are all losing area and thickness but at variable rates, suggesting that local topographic factors affecting solar radiation (shadowing) or preferential snow accumulation areas may play an important role in explaining recent changes. In general, the regional climate conditions (which are the same for all surveyed glaciers) primarily still control the evolution of glaciers in the Pyrenees. Nonetheless, the considerable variability in glacier area and ice thickness losses observed among individual glaciers points to the importance of local topography on glacier mass balance and evolution as well. The results of this study indicate that Pyrenean glaciers are in a clear imbalance with the regional climate and will likely disappear in the next few decades.

### Data Availability Statement

At the time of publication, the database of glacier thickness changes and glacier delimitation in 2011 and 2020 will be available through this URL: https://doi.org/10.5281/zenodo.4756351.

# References

Bahr, D. B., & Radić, V. (2012). Significant contribution to total mass from very small glaciers. *The Cryosphere*, 6(4), 763–770. https://doi.org/10.5194/tc-6-763-2012

Beniston, M., Farinotti, D., Stoffel, M., Andreassen, L. M., Coppola, E., Eckert, N., et al. (2018). The European mountain cryosphere: A review of its current state, trends, and future challenges. *The Cryosphere*, *12*(2), 759–794. https://doi.org/10.5194/tc-12-759-2018

#### Acknowledgments

This work was supported by the Interreg-POCTEFA project OPCC ADAPYR and Spanish Ministry of Economy and Competitiveness project "CGL2017-82216-R." J. Revuelto is supported by the Grant IJC2018-036260-I. I. Vidaller is supported by the Grant FPU18/04978 and is studying the PhD program in the University of Zaragoza. E. Izagirre and I. Rico are supported by the Grant PPGI19/02 (UPV/EHU) and the Consolidated Research Group IT1029-16 (Basque Country Government). E. Berthier acknowledges support from the French Space Agency (CNES).

- Braithwaite, R. J., & Hughes, P. D. (2020). Regional geography of glacier mass balance variability over seven decades 1946–2015 Frontiers in earth science. https://www.frontiersin.org/article/10.3389/feart.2020.00302
- Buisan, S. T., Saz, M. A., & López-Moreno, J. I. (2015). Spatial and temporal variability of winter snow and precipitation days in the western and central Spanish Pyrenees. *International Journal of Climatology*, 35(2), 259–274. https://doi.org/10.1002/joc.3978
- Carturan, L., Filippi, R., Seppi, R., Gabrielli, P., Notarnicola, C., Bertoldi, L., et al. (2013). Area and volume loss of the glaciers in the Ortles-Cevedale group (Eastern Italian Alps): Controls and imbalance of the remaining glaciers. *The Cryosphere*, 7(5), 1339–1359. https:// doi.org/10.5194/tc-7-1339-2013
- Cuadrat, J. M., Valero-Garcés, B., Moreno, A., Verfaillie, D., Galop, D., Rodríguez, E., et al. (2018). El cambio climático en los Pirineos : Impactos, vulnerabilidades y adaptación.
- DeBeer, C. M., & Sharp, M. J. (2009). Topographic influences on recent changes of very small glaciers in the Monashee Mountains, British Columbia, Canada. Journal of Glaciology, 55(192), 691–700. https://doi.org/10.3189/002214309789470851
- Del Rio, M., Rico, I., Serrano, E., & Ramos, J. J. (2014). Applying GPR and laser scanner techniques to monitor the Ossoue Glacier (Pyrenees). Journal of Environmental & Engineering Geophysics, 19, 239–248. https://doi.org/10.2113/JEEG19.4.239
- Fischer, M. (2018). Understanding the response of very small glaciers in the Swiss Alps to climate change. Thesis.
- Fischer, M., Huss, M., Barboux, C., & Hoelzle, M. (2014). The new Swiss Glacier inventory SGI2010: Relevance of using high-resolution source data in areas dominated by very small glaciers. *Arctic Antarctic and Alpine Research*, 46(4), 933–945. https://doi. org/10.1657/1938-4246-46.4.933
- Fischer, M., Huss, M., & Hoelzle, M. (2015). Surface elevation and mass changes of all Swiss glaciers 1980–2010. The Cryosphere, 9(2), 525–540. https://doi.org/10.5194/tc-9-525-2015
- Fischer, M., Huss, M., Kummert, M., & Hoelzle, M. (2016). Application and validation of long-range terrestrial laser scanning to monitor the mass balance of very small glaciers in the Swiss Alps. *The Cryosphere*, *10*(3), 1279–1295. https://doi.org/10.5194/tc-10-1279-2016
- Florentine, C., Harper, J., & Fagre, D. (2020). Parsing complex terrain controls on mountain glacier response to climate forcing. *Global and Planetary Change*, 191, 103209. https://doi.org/10.1016/j.gloplacha.2020.103209
- Florentine, C., Harper, J., Fagre, D., Moore, J., & Peitzsch, E. (2018). Local topography increasingly influences the mass balance of a retreating circu glacier. *The Cryosphere*, 12(6), 2109–2122. https://doi.org/10.5194/tc-12-2109-2018
- Forlani, G., Dall'Asta, E., Diotri, F., Morra di Cella, U., Roncella, R., & Santise, M. (2018). Quality assessment of DSMs produced from UAV flights georeferenced with on-board RTK positioning. *Remote Sensing*, *10*, 311. https://doi.org/10.3390/rs10020311
- Gaffey, C., & Bhardwaj, A. (2020). Applications of unmanned aerial vehicles in cryosphere: Latest advances and prospects. *Remote Sensing*, 12(6), 948. https://doi.org/10.3390/rs12060948

Gascoin, S., & René, P. (2018). Recent evolution of the Vignemale glaciers (2013–2017). https://doi.org/10.3989/pirineos.2018.173004 Girardeau-Montaut, D. (2016). Cloud Compare.

- Grunewald, K., & Scheithauer, J. (2010). Europe's southern most glaciers: Response and adaptation to climate change. Journal of Glaciology, 56(195), 129–142. https://doi.org/10.3189/002214310791190947
- Grunewald, K., Weber, C., Scheithauer, J., & Haubold, F. (2006). Mikrogletscher im Piringebirge (Bulgarien). Gletscherkunde Und Glazialgeologie, 39, 99–114.
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. Quarterly Journal of the Royal Meteorological Society, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Hopkinson, C., & Demuth, M. N. (2006). Using airborne lidar to assess the influence of glacier downwasting on water resources in the Canadian Rocky Mountains. *Canadian Journal of Remote Sensing*, 32(2), 212–222. https://doi.org/10.5589/m06-012
- Hughes, P., & Woodward, J. (2009). Glacial and periglacial environments (pp. 353-383). https://doi.org/10.1093/oso/9780199268030. 003.0024
- Hughes, P. D. (2018). Little ice age glaciers and climate in the mediterranean mountains: A new analysis. *Geographical Research Letters*, 44(1), 15–45. https://doi.org/10.18172/cig.3362
- Hugonnet, R., McNabb, R., Berthier, E., Menounos, B., Nuth, C., Girod, L., et al. (2021). Accelerated global glacier mass loss in the early twenty-first century. *Nature*, 592, 726–731. https://doi.org/10.1038/s41586-021-03436-z
- Huss, M. (2013). Density assumptions for converting geodetic glacier volume change to mass change. *The Cryosphere*, 7(3), 877–887. https://doi.org/10.5194/tc-7-877-2013
- Huss, M., & Fischer, M. (2016). Sensitivity of very small glaciers in the Swiss Alps to future climate change. *Frontiers in earth science*. https://doi.org/10.3389/feart.2016.00034
- Huss, M., & Hock, R. (2018). Global-scale hydrological response to future glacier mass loss. Nature Climate Change, 8, 135–140. https://doi.org/10.1038/s41558-017-0049-x

James, M. (2017). Precision maps and 3-D uncertainty-based topographic change detection with structure-from-motion photogrammetry.

- Jiménez-Vaquero, C. (2016). Cartografiado de la morfología subglaciar de La Maladeta y Aneto mediante georradar. Universitat Politècnica de València.
- Jomelli, V., Chapron, E., Favier, V., Rinterknecht, V., Braucher, R., Tournier, N., et al. (2020). Glacier fluctuations during the Late Glacial and Holocene on the Ariège Valley , northern slope of the Pyrenees and reconstructed climatic conditions. *Mediterranean Geoscience Reviews*, 2(1), 37–51. https://doi.org/10.1007/s42990-020-00018-5
- Leigh, J. R., Stokes, C. R., Carr, R. J., Evans, I. S., Andreassen, L. M., & Evans, D. J. A. (2019). Identifying and mapping very small (<0.5 km<sup>2</sup>) mountain glaciers on coarse to high-resolution imagery. *Journal of Glaciology*, 65(254), 873–888. https://doi.org/10.1017/jog.2019.50
- López-Moreno, J. I. (2005). Recent variations of snowpack depth in the central Spanish Pyrenees. Arctic, Antarctic, and Alpine Research, 37(2), 253–260. https://doi.org/10.1657/1523-0430(2005)037[0253:RVOSD1]2.0.CO;2
- López-Moreno, J. I., Alonso-González, E., Monserrat, O., Del Río, L. M., Otero, J., Lapazaran, J., et al. (2019). Ground-based remote-sensing techniques for diagnosis of the current state and recent evolution of the Monte Perdido Glacier, Spanish Pyrenees. *Journal of Glaciology*, 65(249), 85–100. https://doi.org/10.1017/jog.2018.96
- López-Moreno, J. I., García-Ruiz, J. M., Vicente-Serrano, S. M., Alonso-González, E., Revuelto-Benedí, J., Rico, I., et al. (2020). Critical discussion of: A farewell to glaciers: Ecosystem services loss in the Spanish Pyrenees. *Journal of Environmental Management*, 275, 111247. https://doi.org/10.1016/j.jenvman.2020.111247
- López-Moreno, J. I., Juan, I., Revuelto, J., Alonso-González, E., Sanmiguel-Vallelado, A., Fassnacht, S. R., et al. (2017). Using very longrange terrestrial laser scanner to analyze the temporal consistency of the snowpack distribution in a high mountain environment. *Journal of Mountain Science*, 14(5), 823–842. https://doi.org/10.1007/s11629-016-4086-0
- López-Moreno, J. I., Revuelto, J., Rico, I., Chueca-Cía, J., Julián, A., Serreta, A., et al. (2016). Thinning of the Monte Perdido Glacier in the Spanish Pyrenees since 1981. *The Cryosphere*, *10*(2). 681–694. https://doi.org/10.5194/tc-10-681-2016

- López-Moreno, J. I., Soubeyroux, J. M., Gascoin, S., Alonso-Gonzalez, E., Durán-Gómez, N., Lafaysse, M., & Morin, S. (2020). Long-term trends (1958–2017) in snow cover duration and depth in the Pyrenees. *International Journal of Climatology*, 40(14), 6122–6136.
- Marti, R., Gascoin, S., Houet, T., Laffly, D., René, P., Marti, R., et al. (2015). Evaluation du modèle numérique d'élévation d'un petit glacier de montagne généré à partir d'images stéréoscopiques pléiades : Cas du glacier d'Ossoue, Pyrénées françaises To cite this version : HAL Id. hal-01186226 DE MONTAGNE GENERE A PARTIR D. 57–62.
- Marti, R., Gascoin, S., Houet, T., Ribière, O., Laffly, D., Condom, T., et al. (2015). Evolution of Ossoue Glacier (French Pyrenees) since the end of the Little Ice Age. *The Cryosphere*, 9(5), 1773–1795. https://doi.org/10.5194/tc-9-1773-2015
- Marti, R., Houet, T., & Laffly, D. (2015). The Pyrenean glaciers (South West Europe) in 1850 and 2011: A new crossborder inventory based on aerial images and field surveys. *26th IUGG general assembly*.
- Morán-Tejeda, E., Ceballos, J. L., Peña, K., Lorenzo-Lacruz, J., & López-Moreno, J. I. (2018). Recent evolution and associated hydrological dynamics of a vanishing tropical Andean glacier: Glaciar de Conejeras, Colombia. Hydrology and Earth System Sciences, 22(10), 5445–5461. https://doi.org/10.5194/hess-22-5445-2018
- Parkes, D., & Marzeion, B. (2018). Twentieth-century contribution to sea-level rise from uncharted glaciers. *Nature*, 563(7732), 551–554. https://doi.org/10.1038/s41586-018-0687-9
- Pastor Argüello, F. (2013). Determinación del balance anual de masa Y movimiento del hielo en el glaciar de la maladeta. Año hidrológico 2012–2013, 68. http://www.mapama.gob.es/es/agua/temas/evaluacion-de-los-recursos-hidricos/Informe\_Balance\_Maladeta\_2013\_tcm7-287398.pdf
- Rabatel, A., Castebrunet, H., Favier, V., Nicholson, L., & Kinnard, C. (2011). Glacier changes in the Pascua-Lama region, Chilean Andes (29°S): Recent mass balance and 50 yr surface area variations. *The Cryosphere*, 5, 1029–1041. https://doi.org/10.5194/tc-5-1029-2011
- Rajendra, Y. D., Mehrotra, S. C., Kale, K. V., Manza, R. R., Dhumal, R. K., Nagne, A. D., & Vibhute, A. D. (2014). Evaluation of partially overlapping 3D point cloud's registration by using ICP variant and cloudcompare. *International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, 8(1), 891–897. https://doi.org/10.5194/isprsarchives-XL-8-891-2014
- René, P. (2014). Le suivi des glaciers dans les Pyrénées franÇais. La Météologie, 8. https://doi.org/10.4267/2042/53748
- Revuelto, J., López-Moreno, J. I., & Alonso-González, E. (2021). Light and shadow in mapping alpine snowpack with unmanned aerial vehicles in the absence of ground control points. *Water Resources Research*, *57*. https://doi.org/10.1029/2020WR028980
- Revuelto, J., Zabalza, J., Arguedas, G., Zabalza, J., Arguedas, G., & Vicente-Serrano, S. M. (2014). Mapping the annual evolution of snow depth in a small catchment in the Pyrenees using the long- range terrestrial laser scanning. *Journal of Maps*, 1–393. https://doi.org/10. 1080/17445647.2013.869268
- Rico, I. (2019). Los glaciares de los Pirineos. Estudio glaciológico y dinámica actual en el contexto del cambio global. Tesis Doctoral.

Rico, I., Izagirre, E., Serrano, E., & López-Moreno, J. I. (2017). Superficie glaciar actual en los Pirineos: Una actualización para 2016. Pirineos (Vol. 172, p. e029). https://doi.org/10.3989/Pirineos.2017.172004

- Solomina, O. N., Bradley, R. S., Jomelli, V., Geirsdottir, A., Kaufman, D. S., Koch, J., et al. (2016). Glacier fluctuations during the past 2000 years. *Quaternary Science Reviews*, 149, 61–90. https://doi.org/10.1016/j.quascirev.2016.04.008
- Zemp, M., Frey, H., Gärtner-Roer, I., Nussbaumer, S. U., Hoelzle, M., Paul, F., et al. (2015). Historically unprecedented global glacier decline in the early 21st century. *Journal of Glaciology*, 61(228), 745–762. https://doi.org/10.3189/2015JoG15J017